Author’s response to reviews

Title: A meta-ethnography to understand the experience of living with urinary incontinence: ‘is it just part and parcel of life?’

Authors:

Francine Toye (francine.toye@ouh.nhs.uk)
Karen Barker (karen.barker@ouh.nhs.uk)

Version: 2 Date: 04 Nov 2019

Author’s response to reviews:

Dear Editor

Thank you for considering our manuscript. We are delighted that the reviewers feel that we have adequately responded to their suggestions and that our manuscript makes an important contribution.

We have responded to the editorial comments as follows:

1. We have used correct initials as requested.

2. We have moved the list of abbreviations to before the Declarations and after the Conclusions section.

3. We have removed tracked changes and submitted a non-anonymised clean version.

We look forward to hearing from you in due course.

Kind regards

Fran Toye and Karen Barker